Author's response to reviews

Title: Acute Bilateral Simultaneous Angle Closure Glaucoma after Topiramate Administration

Authors:

Kakarla V Chalam (kchalam@jax.ufl.edu)
Tina Tillis (tina.tillis@jax.ufl.edu)
Farhana Syed (farhana.syed@jax,ufl.edu)
Swati Agarwal (swati.agarwal@jax.ufl.edu)
Vikram S Brar (vikram.brar@jax.ufl.edu)

Version: 5 Date: 18 November 2007

Author's response to reviews: see over
18 November 2007

Editor-in-Chief
Michael Kidd, M.D.

Dear Dr. Kidd

We appreciate the input of the reviewers and yourself regarding our manuscript entitled “Acute bilateral simultaneous angle closure glaucoma after topiramate administration”, manuscript number: 1186447332153318. We have edited our manuscript per the most recent recommendations. The following issues were addressed and specific modifications were made, underlined in the text:

1. Comment was added concerning what we believe is important about this manuscript and how it adds to the current body of literature on page 5 paragraph 2 and page 6.

Looking forward to your positive response.

Sincerely,

K. V. Chalam, MD, PhD, MBA, FACS
Professor and Chairman
Department of Ophthalmology, Health Science Center
University of Florida, Jacksonville Campus
Jacksonville, Florida.